Author's response to reviews

Title: Burden of disease and economic evaluation of healthcare interventions: are we investigating what really matters?

Authors:

Ferrán Catalá-López (ferran_catala@hotmail.com)
Anna García-Altés (annagarciaaltes@gmail.com)
Elena Álvarez-Martín (elena.alvarez.martin@urjc.es)
Ricard Gènova-Maleras (ricard.genova@salud.madrid.org)
Consuelo Morant-Ginestar (consuelo.morant@salud.madrid.org)
Antoni Parada (aparada@aatrm.catsalut.cat)

Version: 5 Date: 21 March 2011

Author's response to reviews:

Dear Editors,

My colleagues and I are extremely happy to submit the revisions of our manuscript.

We appreciated your comments and are confident that the new version satisfies the following editorial requests:

1. On page 8, the word "exemple" (highlighted in yellow) has been changed to "example".

2. On page 11, paragraph 1, last sentence has been revised to read as follows: "We believe these findings do no necessarily imply that resources were allocated inefficiently, but rather raises important questions to keep in mind when funding future economic evaluations."

Finally, we also like to thank the reviewers and editorial staff.

Yours faithfully,

Ferrán Catalá-López (on behalf of all co-authors)